Author's response to reviews

Title: Provider performance in treating poor patients - Factors influencing prescribing practices in Lao PDR: a cross-sectional study

Authors:

Lamphone Syhakhang (syhakhangl@yahoo.com)
Douangdao Soukaloun (dsoukaloun@yahoo.com)
Göran Tomson (goran.tomson@ki.se)
Max Petzold (max.petzold@nhv.se)
Clas Rehnberg (clas.rehnberg@ki.se)
Rolf Wahlström (rolf.wahlstrom@ki.se)

Version: 3 Date: 20 December 2010

Author's response to reviews: see over
Dear, to finalize the manuscript we were asked to include an ethics statement and to document the consent. The following paragraph was added as a last paragraph in the Data analysis section on page 8:

“Ethical clearance was obtained from the National Ethical Committee for Health Research, Ministry of Health, Laos. All participants were informed about the study purpose and gave verbal consent after explanation that all collected information would be handled anonymously.”

We also went back to the formatting checklist once more.

Sincerely, Max Petzold

Nordic School of Public Health